Author's response to reviews

Title: Understanding, comprehensibility and acceptance of an evidence-based consumer information brochure on fall prevention in old age: a focus group study

Authors:

Sabine Lins (Sabine.Lins@gmx.de)
Andrea Icks (Andrea.Icks@uni-duesseldorf.de)
Gabriele Meyer (Gabriele.Meyer@uni-wh.de)

Version: 4 Date: 11 March 2011

Author's response to reviews:

Dear Editor,

Please find attached the re-revised version of our paper “Understanding of an evidence based consumer information on fall prevention in old age: a focus group study” (MS: 8080314124432536).

We would like to thank the reviewers for acknowledging our revision and the editor for his helpful comment.

A point-by-point response is attached and the changes in the manuscript are highlighted in red font.

Yours sincerely,

Gabriele Meyer on behalf of all co-authors